Author’s response to reviews

Title: Estrogen and Soy Isoflavonoids Decrease Sensitivity of Medulloblastoma and Central Nervous System Primitive Neuroectodermal Tumor Cells to Chemotherapeutic Cytotoxicity

Authors:

Scott Belcher (smbelch2@ncsu.edu)
Caleb Burton (cburton@u.northwestern.edu)
Clifford Cookman (cookmacj@mail.uc.edu)
Michelle Kirby (kirby_mich@yahoo.com)
Gabriel Miranda (Gabriel.Miranda@amaisd.org)
Fatima Saeed (fatima.saeed@uc.edu)
Kathleen Wray (wrayk@musc.edu)

Version: 2 Date: 20 Jun 2017

Author’s response to reviews:

The required edits to remove highlighting from the manuscript and include each author's email address were made.